Author's response to reviews

Title: 5-year survival and hospitalization due to stroke recurrence among patients with hemorrhagic or ischemic strokes in Singapore

Authors:

Yan Sun (yan_sun@nhg.com.sg)
Sze Haur Lee (Sze_Haur_Lee@nni.com.sg)
Bee Hoon Heng (Bee_Hoon_Heng@nhg.com.sg)
Vivien S. Chin (vivien_chin@nhg.com.sg)

Version: 2 Date: 31 January 2013

Author's response to reviews: see over
Cover Letter

Dear Editor-in-chief,

Attached please find the revised manuscript of “5-year survival and hospitalization due to stroke recurrence among patients with hemorrhagic or ischemic strokes in Singapore" for consideration in the BMC Neurology.”

The details of the revision according to comments are listed below:

- Ethics Statement and (2) Name of Ethics Committee have been added in the last paragraph of Methods.
- (3) Acknowledgment has been added in the last section of the manuscript.
- Some references have been reformatted according the journal’s requirement.

Should any further requirement, please kindly let us know.

Best regards,

Yan Sun